% paper details
\def\PaperCode{INFO221}
\def\PaperTitle{Application Software Development}

% semester, and year
\def\Semester{2}
\def\Year{2014}

% lecture number and title
\def\LectureNumber{21}
\def\LectureTitle{Collections Part 2}

% lecturer and institute
\def\Lecturer{Mark George {\footnotesize (\textsf{mgeorge@infoscience.otago.ac.nz})}}
\def\Institute{\footnotesize Department of Information Science, University of Otago}

% the details for \maketitle
\author{\emph{Lecturer}: \Lecturer \\ \Institute}
\title{\emph{\PaperCode}:~  \PaperTitle \\ \vspace{10pt} \emph{Lecture \LectureNumber}: \LectureTitle}
\date{\footnotesize Semester \Semester, \Year}


% details to be embedded in the PDF document properties
\hypersetup{
	pdftitle={Lecture \LectureNumber:  \LectureTitle},
	pdfsubject={\PaperCode: \PaperTitle},
	pdfauthor={\Lecturer}
}
